Author’s response to reviews

Title: Recovery-focused mental health care planning and co-ordination in acute inpatient mental health settings: a cross national comparative mixed methods study

Authors:

Michael Coffey (m.j.coffey@swansea.ac.uk)
Ben Hannigan (hanniganb@cardiff.ac.uk)
Sally Barlow (sally.barlow.1@city.ac.uk)
Martin Cartwright (martin.cartwright.1@city.ac.uk)
Rachel Cohen (rachel.cohen@bristol.ac.uk)
Alison Faulkner (alison.faulkner2@btinternet.com)
Aled Jones (jonesa97@cardiff.ac.uk)
Alan Simpson (a.simpson@city.ac.uk)

Version: 1 Date: 24 Aug 2018

Author’s response to reviews:

1) Please mention and cite in the beginning of the manuscript that this manuscript is part of the larger Simpson et al.2017 publication in Health Services and Delivery Research.

RESPONSE: we have included explicit reference to the larger study in the background section at the beginning of the paper. We have inserted this where we felt it worked within the existing text rather than as a standalone statement but we are happy to be advised if this does not meet your request.

2) Please edit Figure formatting:

RESPONSE: All figures are submitted as separate files and a separate list of boxes, tables and figures are now given at the end of the manuscript. Again happy to be advised if we have not fully understood the direction here.
3) If you do include Supplementary Material please make sure you reference it in the body of the manuscript and list the following information in a separate section of the manuscript text.

RESPONSE: We had included a supplementary file and reference to it in the paper but have now added this to the list at the end of the paper.